Author’s response to reviews

Title: Systematic Review of Methods for Evaluating Healthcare Research Economic Impact

Authors:

Bahareh Yazdizadeh (baharehyazdizadeh@yahoo.com)
Reza Majdzadeh (rezamajd@tums.ac.ir)
Hojat Salmasian (hojjat_salmasian@yahoo.com)

Version: 6 Date: 22 February 2010

Author’s response to reviews:

Dear Editor-in-chief,

Thank so much for accept to publish our paper in your valuable journal. We are pride oneself on it.

There is one point about affiliation of authors. For Bahareh Yazdizadeh it’s important that “Department of Biostatistics and Epidemiology, School of Public Health” be the first affiliation and For Reza Majdzadeh, it’s is important that the “Knowledge Utilization Research Center “be the first affiliation.Both of these are from “Tehran University of Medical Science”.

Sincerely Yours,